



# **A Review on Development of Industrial Solid Waste in Tunnel Grouting Materials: Feasibility, Performance, and Prospects**

Bolin Jiang <sup>1,2,3</sup>, Mengjun Wu <sup>2,3</sup>, Shanshan Wu <sup>3,4</sup>, Aichen Zheng <sup>1,3</sup> and Shiyong He <sup>3,\*</sup>

- <sup>1</sup> Chongqing Vocational Institute of Engineering, Chongqing 402260, China; bolinjiang@cqvie.edu.cn (B.J.); aichenzheng@cqvie.edu.cn (A.Z.)
- <sup>2</sup> China Merchants Chongqing Communications Technology Research & Design Institute Co., Ltd., Chongqing 400067, China; mengjuwu@163.com
- <sup>3</sup> School of Civil Engineering, Chongqing Jiaotong University, Chongqing 400074, China; shanshanwu84@163.com
- <sup>4</sup> Chongqing Vocational College of Public Transportation, Chongqing 402247, China
- \* Correspondence: heshiyong@cqjtu.edu.cn

Abstract: With rapid infrastructure development worldwide, the generation of industrial solid waste (ISW) has substantially increased, causing resource wastage and environmental pollution. Meanwhile, tunnel engineering requires large quantities of grouting material for ground treatment and consolidation. Using ISW as a component in tunnel grouts provides a sustainable solution to both issues. This paper presented a comprehensive review of the recent advancements in tunnel grouting materials using ISW, focusing on their feasibility, mechanical characteristics, and future development directions. Initially, the concept and classification of ISW were introduced, examining its feasibility and advantages as grouting materials in tunnels. Subsequently, various performances of ISW in tunnel grouting materials were summarized to explore the factors influencing mechanical strength, fluidity, durability, and microstructure characteristics. Simultaneously, this review analyzed current research trends and outlines future development directions. Major challenges, including quality assurance, environmental risks, and lack of standardized specifications, are discussed. Future research directions, including multifunctional grouts, integrated waste utilization, and advanced characterization techniques, are suggested to further advance this field. These findings provided useful insights for the continued development of high-performance and environmentally friendly ISW-based grouting materials.

Keywords: literature review; industrial solid waste; tunnel grouting materials; performance; applications

# 1. Introduction

In recent years, infrastructure investment around the world has accelerated, resulting in a surge of tunnel construction projects for transportation, hydropower, and urban development [1–4]. Tunneling often takes place in soft ground or fractured rock, requiring large quantities of grouting materials for ground improvement and water inflow control [5–8]. Traditional cement and sodium silicate-based grouts used in tunnels incur high costs and substantial carbon emissions. Meanwhile, rapid industrialization and urbanization have led to a massive generation of industrial solid waste (ISW), including slag, fly ash, steel slag, red mud, etc. [9–12]. Annual global ISW production continues to increase, as shown in Figure 1. The annual production of ISW worldwide continues to increase, and China produces more than 10 billion tons of ISW annually [13–15]. Stockpiling these ISWs wastes resources and poses contamination risks to soil and groundwater. The requirement of reasonable disposal and the recycling of ISW is urgent in the world. Some studies reveal that ISW has the potential and feasibility to be used as grouting material (double-liquid grouts and single-liquid grouts) in tunnel engineering [16–21]. Therefore, increasing research efforts have focused on recycling ISW as substitute constituents in cementitious grouting



Citation: Jiang, B.; Wu, M.; Wu, S.; Zheng, A.; He, S. A Review on Development of Industrial Solid Waste in Tunnel Grouting Materials: Feasibility, Performance, and Prospects. *Materials* **2023**, *16*, 6848. https://doi.org/10.3390/ ma16216848

Academic Editor: Francisco Agrela

Received: 15 September 2023 Revised: 13 October 2023 Accepted: 18 October 2023 Published: 25 October 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



materials; the details are shown in Figure 2. This emerging approach provides a promising solution to cut tunnel costs while reducing environmental pollution from ISW.

Figure 1. Annual output of solid waste in China.



Figure 2. Composition of grouting material.

Compared to traditional grouting materials, ISW grouts exhibit certain advantages, such as high early strength, good impermeability, and improved durability [22–25]. Specific project examples are shown in Table 1. The complex chemical and mineral components in ISW can participate in pozzolanic reactions and generate cementitious gels. When

rationally designed, ISW grouts can achieve equivalent or superior performance to conventional grouts [26]. However, utilizing ISW in grouting materials also faces some technical challenges. ISW comprises waste and by-products generated during industrial production processes, which contain various chemical components, resulting in different physical characteristics; the details are shown in Table 2. Due to the variability in ISW properties, obtaining a consistent grout quality that meets engineering requirements is difficult to guarantee; the details are shown in Figure 3 [27–29]. Furthermore, standardized codes and specifications for ISW grout production and application are still lacking. Zhang et al. [30] conducted a review study to examine the research progress of ISW as cementitious materials, highlighting various utilization methods and their application effects. As technology advanced and environmental concerns grew, research on ISW as grouting material gradually increased, leading to its practical application in engineering projects.

Table 1. The summary of case studies on the application of grouting materials.

Application Area	Engineering Environment	Material Composition	Reasons For Selection	Ref.
Changsha, Hunan Province, China	Karst ground	Sodium bentonite, lignin, meta-aluminate, cement	Better stability	[31]
Qingdao, Shandong Province, China	Marine environment	OPC, sodium silicate	Higher erosion resistance and better strength	[32]
Changsha, Hunan Province, China	Loose ground	Cement, clay, sodium bentonite, gypsum	Better stability and permeability	[33]
Wuhan, Hubei Province, China	High-pressure, water-rich strata	OPC, sand, fly ash, silica fume, steel slag, sodium bentonite	Better stability	[34]
Wuhan, Hubei Province, China	Under river	OPC, steel slag, fly ash, slag, metakaolin, sodium silicate,	Better corrosion resistance	[35]
Nanjing, Jiangsu Province, China	Soft soil strata	OPC, fly ash, hydrated lime, sand, clay, sodium bentonite,	Good workability and corrosion resistance	[36]
Wuhan, Hubei Province, China	Water-rich strata	OPC, sand, fly ash, silica fume, SPs, HEC	Good flow and strength properties	[37]
Xi'an, Shaanxi Province, China	Loess ground	OPC, sand, fly ash, sodium bentonite, micro-expansive agents	Higher strength and good workability	[38]
Beijing, China	Water-rich strata	OPC, fly ash, slag, sodium silicate, sodium bentonite, retarder agents	Better liquidity performance	[39]
Beijing, China	Sulfate-rich environment	Cement, fly ash, slag, sodium silicate, sodium bentonite	Good flow and strength properties	[40]
Chengdu, Sichuan Province, China	Complex ground	OPC, CSA, silica fume, early strength agent	High early strength and better durability	[41]

Table 2. Specific chemical composition of solid waste.

Species	Fe <sub>2</sub> O <sub>3</sub> /%	FeO /%	Al <sub>2</sub> O <sub>3</sub> /%	CaO /%	MgO /%	SiO <sub>2</sub> /%	MnO /%
Cyclone ash	66.45	10.42	3.21	1.74	0.32	4.91	0.12
Sintered dust	64.89	2.8	2.9	14.2	3.05	6.35	0.23
Converter Dephosphrized steel slag	3.83	12.52	3.55	39.66	3.02	15.25	6.08
Species	TiO <sub>2</sub> /%	Na <sub>2</sub> O /%	K2O /%	ZnO /%	Cl /%	P <sub>2</sub> O <sub>5</sub> /%	Other /%
Cyclone ash	0.08	0.16	0.15	0.31	0.59	0.05	11.49
Sintered dust	0.09	0.06	0.3	0.03	0.28	0.06	4.76
Converter Dephosphrized steel slag	2.03	0.3	0.36	0.014	-	6.58	6.8

The potential and suitability of different ISW types in the production of grouting materials were explored by investigating their physical, chemical, and mechanical properties. For instance, waste slag, fly ash, and steel slag have been extensively studied and applied in grouting material preparation; the details are shown in Figure 4 [42–47]. Furthermore, the development of chemical grouting technology has opened up more possibilities for ISW utilization. Sun et al. [48] investigated the use of chemical grouting technology to convert waste steel slag into high-performance grouting materials. By adjusting the chemical formula and process parameters, they successfully transformed waste steel slag into grouting materials with exceptional properties; a range of 20–50% ISW content is proposed, which seems to be optimal for balancing the strength gain with the loss of workability and demonstrating the application potential of ISW in the field of grouting materials. Therefore, the research on utilizing ISW as a tunnel grouting material has undergone a significant and meaningful development process.



Figure 3. The four hazards of solid waste.





There is an increasing focus on the impact of ISW on the mechanical properties of grouting materials, prompting researchers to delve deeper by experimental and microscopic methods. Wang et al. [49] conducted experiments and simulation studies to investigate the influence of various waste components on grouting material performance, providing valuable insights for engineering applications. Ren et al. [50] examined the effects of different ISW components on the mechanical properties of grouting materials. They selected diverse ISWs, including waste slag, fly ash, and steel slag, and mixed them with cement to prepare a series of grouting materials. In a separate study, Sha et al. [51] focused on the influence of steel slag particle-size distribution on the mechanical properties of grouting materials. They prepared a range of grouting materials by sieving and classifying steel slag with varying particle size distributions. Additionally, Mirza et al. [52] investigated

the effect of curing conditions on the mechanical properties of fly ash-containing grouting materials. They examined the impact of curing conditions on the strength, toughness, and deformation properties of these materials through compression and tensile tests. Therefore, a better understanding of the feasibility and optimization methods for utilizing ISW as a tunnel grouting material can be achieved, by studying the effects of different waste compositions, particle size distributions, and curing conditions on the mechanical properties of grouting materials.

The authors of this study concentrated on publications containing keywords such as "industrial solid waste", "tunnel grouting materials", "performance enhancement", "process optimization", "durability", and "fluidity". Various scientific databases, including "Web of Science", "Google Scholar", "Science Direct", "Science Citation Index", and "Scopus", were utilized for the literature search. Based on this approach, a literature review of over 130 articles from the past decade was conducted.

This paper aims to provide a comprehensive overview of the recent studies on ISW grouting materials for tunnel engineering, analyze key factors influencing their properties, summarize the major issues and challenges, and discuss future research directions. It is divided into four main sections. Section 2 examines the feasibility of using different ISWs in tunnel grouting. Section 3 discusses the effects of ISW on key grout properties, including strength, fluidity, and durability. Section 4 identifies major challenges and suggests future work to realize the potential of ISW grouts. Section 5 provides the overall conclusions.

#### 2. Feasibility of Industrial Solid Waste as Grouting Material

ISW primarily consists of waste slag, fly ash, and other ISW. The usage of ISW presents significant challenges due to its large output, varied types, and complex composition. A wide range of industrial wastes have been investigated for their viability as substitute materials in tunnel grouting formulations [53]. This section summarizes recent studies on the feasibility of incorporating different ISWs as partial replacements for cement or aggregates in grout mixtures. The effects of ISW composition and proportions on the properties of fresh and hardened grouts are also discussed.

## 2.1. Slag

Slag is a by-product of iron and steel smelting processes. Research shows that slag powder can be used to partly replace cement in tunnel grouts. Lin et al. [54] found that slag undergoes hydration reactions when activated by cement and lime; the details are shown in Figure 5. The reaction products include calcium silicate hydrates (C-S-H) and calcium aluminate hydrates (C-A-H), which enhance strength. Li et al. [55] discovered that the compressive strength of the slag reached 20 MPa after the treatment, and the slag was suitable as a cementing material to replace a part of Portland cement. Prentice et al. [56] addressed the issues of a prolonged setting time and low early strength of (low clinker) slag cement by incorporating admixtures and implementing a compound preparation scheme. Through X-ray diffraction (XRD) testing and analysis, Yuyou, Zengdi, Xiangqian, and Haijun [40] explored the mechanism of tricalcium silicate in cement minerals to enhance the setting time and early compressive strength. Li et al. [57] optimized the particle size distribution of slag powder to achieve a 7-day grout strength exceeding 6 MPa with 30% slag incorporation.

Steel slag is a by-product of steel manufacturing. Its hydraulic properties allow it to be used in grouting. Ghouleh et al. [58] showed that hardened and graded steel slag aggregates improve grout stability. Up to 70% of natural aggregates can be substituted by steel slag. However, a higher steel slag content increases shrinkage. Li et al. [59] developed a cement-based composite material using ISW (steel slag) and sulphoaluminate cement clinker. This material leverages the cementitious activity of the silicon-aluminum component of steel slag, resulting in improved fluidity, stability, and resistance to ion erosion of the grouting material. André et al. [60] prepared grouts using ground granulated



blast furnace slag and found that finely ground steel slag accelerates C-S-H gel formation, enabling early strength development.

Figure 5. Schematic diagram of hydration reaction.

Gencel et al. [61] investigated the use of waste steel slag as a cementitious material in cement mortar. They analyzed the composition and structural characteristics of ISW and evaluated the active characteristics, curing degree, and performance of the activated cementitious body. Sun and Chen [62] studied the effects of steel slag fineness and the silicate modulus of sodium silicate solution on grout performance. A slag fineness of 7000 cm<sup>2</sup>/g and a silicate modulus of 2.5 were optimal. Yu et al. [63] suggested that a steel slag content between 30–50% in grouts provided favorable rheology, setting time, and strength. Grade S95 GGBS (ground granulated blast slag) and class F fly ash were used as solid precursors to produce the AAISW-based two-component grout in this study. GGBS particles are irregular and angular, while FA particles are spherical in different sizes, as shown in Figure 6.



#### (a)

(b)

Figure 6. SEM images of solid precursors: (a) GGBS and (b) FA.

# 2.2. Fly Ash

Fly ash is a fine powder produced from the combustion of coal. Fly ash exhibits strong pozzolanic reactivity and has been widely studied for grout applications. Pei et al. [64]

found that class F fly ash can serve as a supplementary cementing material in grouting materials. The results revealed that incorporating 5% fly ash into the grouting material increased its compressive strength by more than 22% compared to the original grouting material. This enhancement is attributed to the filling of pores between aggregates by the calcium carbonate produced during the carbonation process of fly ash, facilitating the formation of a dense structure; the details are shown in Figure 7. Trimurtiningrum and Ekaputri [65] developed a grout mixture with a 70% class C fly ash replacement of cement. With a 10% gypsum addition, the grout achieved a 28-day strength of over 100 MPa. Li et al. [66] found that class C fly ash refinement is key to improving grout properties. Fly ash participation in pozzolanic reactions also enhances long-term strength. The optimal fly ash content is around 30% to 50% for strength and fluidity considerations.



Figure 7. Schematic diagram of acidification reaction.

#### 2.3. Other ISWs

Apart from the aforementioned categories, other ISWs, including red mud, carbide slag, and water treatment residuals, have also been studied for grout production. Composite blends using two or more ISWs can further improve the grout properties compared to single admixtures. Huang et al. [67] investigated the influence of the slag content, the steel slag content and fineness on the autogenous shrinkage of Portland cement. The results indicated that the hydration activity of steel slag plays a crucial role in determining the autogenous shrinkage of the cement. Higher hydration activity leads to greater autogenous shrinkage. Cui et al. [68] employed grouting reinforcement technology and used sulphoaluminate cement as the primary material mixed with prepared talc-like substances. Deng and Zheng [69] utilized zinc slag as a raw material for grouting material derived from ISW. Zhang et al. [70] investigated the chemical structure, surface characteristics, and physical and mechanical properties of sulphoaluminate cement-based grouting materials. They enhanced the mechanical properties of these grouting materials by incorporating an appropriate amount of ISW admixture and water-soluble polyurethane.

Guo et al. [71] employed a significant amount of fly ash, coal gangue, a suitable quantity of desulfurization gypsum, and an alkali activator, along with a small amount of admixture, to prepare coal-based ISW grouting material. Additionally, they utilized response surface analysis to determine the optimal dosage for achieving the desirable fluidity, gel time, and other properties. Overall, an ISW content of around 40% is appropriate for balancing strength enhancement versus workability reduction. The effectiveness also depends on the ISW characteristics and activation methods.

Table 3 shows the feasibility results of different ISWs as grouting materials. It can be seen that using solid waste as grouting material can effectively reduce construction costs while facilitating the safe and efficient recycling of solid waste in engineering applications. Furthermore, treating ISW requires careful consideration of grouting performances in tunnel engineering, as well as the potential environmental pollution resulting from solid waste. The usage of ISW holds significant practical value and contributes to environmental protection, human health preservation, and sustainable development promotion.

ISW Type	Procedure	Improved Performance	Feasibility
Slag	Composite addition with admixture	Enhanced strength performance	Feasible
Steel slag	Optimize particle size distribution	Fluidity, stability, and resistance to ion erosion	Feasible
Fly ash	The amount of incorporation in the appropriate range	Increase compressive strength	Feasible
Other ISWs (including red mud, carbide slag, water treatment residuals, etc.)	Two or more than two types of composite incorporation	Improve fluidity, gelation time, and other working properties.	Feasible

 Table 3. Feasibility summary of various ISW materials as grouting materials.

# 3. Effect of Industrial Solid Waste on Performance of Grouting Material

With the rapid development of society, a generation of various industrial solid wastes (ISWs), including steel slag, slag, and fly ash, has surged. The large-scale stockpiling of these ISWs not only occupies land resources but also causes environmental pollution around the storage sites. Therefore, the comprehensive utilization of these resources is imperative. Currently, the incorporation of ISWs into building materials has become a research hotspot. Numerous studies have proven that the addition of appropriate amounts of ISWs to cement-based grouting materials can significantly reduce raw material costs while delivering enhanced mechanical performance [72]. This section examines how the incorporation of ISW influences the various grout properties essential for engineering performance, including strength, fluidity, impermeability, volume stability, and durability. The effects of the ISW type and dosage are analyzed based on recent studies.

## 3.1. Mechanical Strength of Grouting Material

Based on previous research, it summarizes the effects of different ISWs on the mechanical strength of cementitious grouting materials and discusses the mechanisms behind ISW incorporation to provide a theoretical basis for the resource utilization of ISWs.

#### 3.1.1. Industrial Solid Waste Incorporation and Process Optimization

The strength development in ISW grouts is linked to the pozzolanic reactivity of ISW components [73]. Si, Al, and Fe elements in ISW source materials can react with the Ca(OH)<sub>2</sub> released during cement hydration to form additional strength-enhancing products. The porous structure of ISW particles also serves as nucleation sites, refining pore structure and increasing density.

Common ISWs, such as slag, fly ash, and steel slag, are widely used in cementbased grouting materials. For instance, the combination of slag powder and cement can produce materials with high early strength for full tailings filling, while the mixture of steel slag slurry and cement exhibits significantly improved compressive strength after carbonation [74]. The proportional incorporation of Bayer red mud, carbide slag, and desulfurized gypsum continuously enhances material strength during the carbonation process due to the gradual formation of calcium carbonate. At the same time, the Bayer process will generate the following types of industrial waste (including arsenic waste, silica ferrous slag, slurry wastewater, etc.), and to minimize the environmental impact and waste of resources, the relevant industrial waste needs to be treated and managed appropriately.

Cao et al. [75] combined slag powder and P.O. 42.5 cement to develop a cementitious material with specific early strength for full tailings filling. The composition consisted of 60% slag, 15% cement, 10% hemihydrate gypsum and desulfurization gypsum, and 5% lime. Xue et al. [76] utilized slag powder, along with clinker and gypsum, to develop a cementitious material for the purpose of whole tailings filling. This material exhibited a remarkable strength that was 2.7 times higher than that of cement-based filler in similar conditions. Furthermore, it proved to be a cost-effective alternative, with a significantly lower cost ranging from 40% to 45% of traditional cement. Sha et al. [77] combined pure steel slag slurry and Portland cement to form a steel slag mixture. After carbonization, the

compressive strength of the mixture significantly improved. Deng et al. [78] developed a graded tailings filling cemented body using lead-zinc smelting slag, clinker, sodium silicate, gypsum, and other activating materials, with a cement-sand ratio of 1:6. The strength of the cured samples after 3 days reached 2.68 MPa, while after 28 days, it reached 3.97 MPa, representing a significant 21% increase compared to traditional P.O. 42.5 cement. Li et al. [79] developed a dual-liquid grouting material, incorporating industrial waste residues such as steel slag, slag, fly ash, and water glass. The material exhibited a gel time of 3–300 s and a strength of 15–25 MPa. This material significantly enhances the utilization of industrial waste residues and reduces cement grouting consumption. Lahalle et al. [80] developed an alkali-activated polymer double-liquid grouting material using sodium silicate-activated slag and metakaolin. They proposed that when the mass ratio of slag to metakaolin was 1:2, and the volume content of glass was 20%, the compressive strength at 3 days and 28 days reached 7.23 MPa and 17.26 MPa, respectively. Bai et al. [81] prepared a fly ash-based grouting material with an initial setting time of 13 min, a final setting time of 20 min, and a compressive strength of 104.5 MPa. Wan et al. [82] addressed the slow hydration rate of a steel slag–slag–barium slag composite grouting material by utilizing desulfurization gypsum as an activator. The results showed a 21% increase in the 28-day strength. Sha et al. [83] prepared a new type of cement-fly ash-modified sodium silicate grouting material. The results showed that the initial setting time increased, and the final setting time decreased with an increase in the volume fraction of modified sodium silicate. The gel time was proportional to the mass fraction of fly ash in the powder, while the slurry's fluidity was proportional to the volume fraction of modified sodium silicate and the mass fraction of fly ash in the powder.

Further enhancement of grouting material performance can be achieved by optimizing the addition amounts, combinational types, and process parameters of ISWs. Zhu et al. [84] prepared cementitious materials using slag and fly ash as raw materials under alkali activation. The findings revealed that the compressive strength initially increased and then decreased with a decreasing fly ash content when 3% sodium hydroxide (by mass) was used as an alkali activator and the water-to-solid ratio was 0.4. Juan-hong et al. [85] prepared a grouting material by mixing red mud, carbide slag, and desulfurized gypsum at a mass ratio of 74:11:15. The results demonstrated that the content of calcium carbonate in the material increased with carbonization time, leading to a gradual increase in compressive strength. Zhang et al. [86] developed a novel double-liquid grouting material with an adjustable gel time, high consolidation strength, and a stone rate of up to 100%. This material was produced by combining industrial waste residue with Portland cement clinker, optimizing particle size distribution, and alkaline-activating cementitious activity.

Carvalho et al. [87] investigated the impact of steel slag replacement on the performance of composite filling cementitious materials. The findings revealed a pattern of initial shrinkage followed by expansion as the steel slag replacement amount increased, with shrinkage reaching its lowest point at 83%. Chao et al. [88] conducted a comprehensive tailings filling test using a cost-effective grouting material they developed, primarily composed of fly ash, slag powder, carbide slag, and a composite activator. The results demonstrated that with a cementing agent-to-tailings ratio of 1:4–1:8 and a slurry concentration of 67–70%, the 28-day compressive strength ranged from 0.68–2.63 MPa. Ehsani et al. [89] examined the water absorption, fluidity, and compressive strength of a novel fly ash-cement grouting material and determined that optimal performance was achieved with a fly ash ratio of 17–20% and a water–cement ratio of 0.65–0.7. Rehman et al. [90] examined the performance of a steel slag-modified silicate-sodium silicate double-liquid grouting composite material. Their research determined that the optimal water–binder ratio ranged from 0.6–0.8, the ideal volume ratio of sodium silicate to the slurry was 1:4-1:6, and the steel slag content constituted 50-80% of the cement mass. After 3 days of curing, the material with the ratios achieved an early strength exceeding 40 MPa. To enhance the comprehensive utilization of solid waste, Xue et al. [91] conducted an optimization test on subsection iron removal from steel slag and investigated the influence of steel slag powder on the compressive

strength of the grouting material. The findings revealed that through sectional magnetic separation, high-performance steel slag powder with less than a 0.5% metal iron content could be obtained. Additionally, when the specific surface area of the steel slag powder was 1:2.5, the material achieved improved 3-day and 28-day strength simultaneously.

Gopinathan and Anand [92] investigated the mechanical properties of grouting materials with a high content of slag powder. The results demonstrated that as the slag incorporation rate increased, the slurry density gradually increased, porosity decreased, and strength increased. Additionally, under low-temperature conditions, the high-content slag powder grouting material exhibited an improvement in compressive strength by 18–27% compared to conventional grouting materials.

Based on a comprehensive analysis of steel slag powder characteristics, Liu and Guo [93] incorporated it into a steel slag-cement and steel slag-mineral powder-fly ash composite system. The research indicated that the optimal replacement rate of steel slag powder in cement was 15%, and the optimum ratio of steel slag powder to mineral powder-fly ash was 1:2:1. Furthermore, the use of Na<sub>2</sub>SO<sub>4</sub> as an activator in the steel slag powder-cement composite system effectively improved the early strength of cementitious materials.

## 3.1.2. Mechanisms of Performance Enhancement

The incorporation of ISWs promotes cement hydration reactions and generates more hydration products, such as ettringite and C-S-H gel. For example, the addition of steel slag and slag induces hydration reactions when activated by cement [94]. The initial hydration of steel slag promotes the depolymerization of steel slag, which subsequently reacts with cement to generate ettringite network structures that are subsequently filled by C-S-H gel, thereby increasing the density of the hardened paste system.

Li et al. [95], guided by grouting engineering performance, primarily utilizes red mud from ISW as the main raw material to prepare a red mud-based grouting material in collaboration with various solid waste materials. The occurrence of free water, the hydration exothermic law, hydration kinetics characteristics, alkaline components, and the heavy metal content of the slurry are analyzed. The hydration mechanics characteristics of the red mud-based grouting material are determined, and its hydration mechanism is revealed. The hydration dynamics characteristics are analyzed using the Krstulovic–Dabic model.

Zhang et al. [96] found that different reaction degrees and sizes of embedded microstructures were observed in fly ash particles at different ages, adversely affecting the mechanical properties of the materials. The activation degree of fly ash gradually increased with age, contributing continuously to the strength development in the later stages. This is due to the fact that with the increase of age, the active substances in the fly ash (the activated substances refer to the components in the fly ash that can react with water to generate cementitious substances, which play an important role in promoting the development of the later strength of the material) will be gradually depleted, and the degree of activation will be gradually reduced, which will adversely affect the development of the later strength.

Liu et al. [97] investigated the hydration reaction process of a clinker-free cementitious system comprising steel slag, slag, and gypsum. The results revealed that an increase in the specific surface area of steel slag powder enhanced the compressive strength of non-clinker concrete but reduced its slump. The network structure formed by ettringite during the early stages played a vital role in the material's early compressive strength, while the hydration products and C-S-H gel continuously filled the gaps in the hardened paste, enhancing the overall compressive strength of the material.

From the above studies, it is easy to see that the use of ISW as grouting material can not only realize the comprehensive utilization of resources but also have significant economic benefits. However, although the existing studies on various ISW-modification mechanisms can provide some theoretical references for practical projects, the systematic selection of ISW materials and optimal dosage to produce low-cost and high-performance grouting materials requires further exploration.

## 3.2. Fluidity of Grouting Material

Grouting applications in tunnels, foundations, and structural repairs demand controllable fluidity and strength [98]. High fluidity is required during grout injection to achieve good penetration and filling capacity. ISW particles can improve particle packing and lubrication between cement grains, enhancing grout fluidity [99]. However, an excessive ISW dosage can sharply increase viscosity due to a higher specific surface. This section comprehensively examines the interconnected factors influencing the fluidity of grouting material.

The effects of alkali metal cations ( $K^+$ ,  $Na^+$ ) on slurry rheology and setting behaviors were evaluated. Li et al. [100] investigated the slurry viscosity and setting time of grouting materials mixed with different alkali metal cations using Fourier transform infrared spectroscopy (FTIR) and viscosity tests. The results indicate that when the ratio of alkali metal cation ( $K^+$ ) to silicon ion is less than 1.5, the initial setting time of alkali slag grouting material can be effectively prolonged. Additionally, the addition of an appropriate amount of  $K^+$  can effectively reduce the viscosity of the solution and improve the material's fluidity.

The alkali type/content regulated the viscosity and setting, enhancing workability below a threshold. Fly ash and slag modulated the workability and accelerated the strength gain in optimal dosages. Moghadam et al. [101] activated coal gangue, steel slag, and fly ash using NaOH and Na<sub>2</sub>SO<sub>4</sub> to prepare cement-free, controllable, low-strength grouting material. When the mass fraction of fly ash is below 40%, the workability (fluidity) of the grouting material meets the ACI (American Concrete Institute) specifications. The phase composition of reaction products in the routing material system is influenced by the mass ratio of steel slag to fly ash. Xu et al. [102] studied the influence of fly ash on the working performance of grouting materials. The results demonstrate that the addition of fly ash significantly improves the fluidity of the materials, and the fluidity increases gradually with the increasing fly ash content up to 40%, resulting in good working performance. ElKhatib et al. [103] successfully prepared various compaction grouting materials for tunnel grouting by utilizing steel slag, fly ash, and other raw materials based on the activity characteristics of steel slag. The results show that the addition of steel slag improves the fluidity of the grouting material, reduces shrinkage, accelerates setting time, and significantly enhances the 7-day and 28-day strength.

Xiang et al. [104] investigated the effect of large amounts of mineral admixtures, such as fly ash and mineral powder, on the performance of grouting materials. The results indicate that a material with a composition of 30% fly ash and 10% mineral powder achieves optimal fluidity. Wu et al. [105] conducted an experimental study on the basic performance of solid waste grouting filling material using rubber particles, fly ash, and clay as the main materials. The results show that a grouting slurry with a fluidity of 293 mm, a 28-day compressive strength of 11.7 MPa, and an impermeability pressure of 0.8 MPa can be achieved when the content of rubber particles is 20%, fly ash is 65%, and clay is 15%.

In summary, the above studies have elucidated the rheological properties, microstructureproperty linkages, and composition-property linkages of ISW grouting materials, providing theoretical guidance for the design of sustainable materials. However, due to the limitations of microscopic equipment and numerical simulation computational capabilities, the mechanistic explanations of the above phenomena have not yet gained a unified perception.

#### 3.3. Durability of Grouting Material

Grouting materials are widely used in tunnel engineering to consolidate surrounding rock and soil, control groundwater infiltration, and prevent corrosion. However, grouting materials often face harsh and complex service environments that can impact their durability and performance. This section categorizes and summarizes the recent studies on various factors influencing the durability of grouting materials, including volume stability, water corrosion resistance, impermeability, sulfate attack resistance, and freeze–thaw resistance.

## 3.3.1. Volume Stability

Volume stability is critical for grouting materials to avoid excessive shrinkage and cracking. Given that the service environment of grouting material is dynamic and intricate, the volume stability of the grout body is closely associated with the engineering's service life. Excessive shrinkage of the grout can lead to crack formation, which in turn can cause leakage and intensify the erosion caused by corrosive ions. He et al. [106] found that high early strength cement grouts exhibit micro-expansion without compromising strength. This expansibility tends to stabilize within the initial 7 days without compromising the subsequent strength of the grout body. Furthermore, micro-expansion also contributes to improving the compactness of hardened slurry. Li and Hao [107] showed that the early strength agent increased the early chemical shrinkage of the double-liquid sulphoaluminate grouting material. Under standard oxidation conditions, the grout experienced volume expansion; however, under dry air curing, it exhibited volume shrinkage. Borçato et al. [108] determined that the shrinkage rate decreases with increasing metakaolin content in geopolymer grouts. The results demonstrated that the shrinkage rate of the grout body decreased with increasing high terrestrial content. However, when the content exceeded 30%, the effect began to weaken. The application of industrial solid waste flue gas desulfurization gypsum (FGDG) in magnesium oxy-sulfate cement (MOSC) was investigated by Li et al. [109]. The results showed that the volume stability of the mixture with the optimum admixture was better at the curing temperature of 40 °C, and its 28-day compressive strength could reach 56.6 MPa, which was 12.5% higher than that of the curing condition at 20 °C, and the softening coefficient could reach 0.99. Meanwhile, it was also confirmed that more hydration products  $(5Mg(OH)_2 \cdot MgSO_4 \cdot 7H_2O)$  were produced in the mixture under the curing condition of 40 °C through SEM and XRD, and its structure is denser.

## 3.3.2. Water Corrosion Resistance

Groundwater can cause the gel product in the hardened grout stone body to decompose and dissolve through physical and chemical actions, resulting in a decrease in the mechanical properties of the stone body [110]. This phenomenon is known as water corrosion. Therefore, flowing water can cause dissolution and deterioration of grout gel products and strength loss. Bentz and Garboczi [111] identified calcium hydroxide as vulnerable to leaching and dissolution under the influence of flowing water. This is especially true under conditions of high permeability and high-water pressure in hardened cement paste, where water can infiltrate the stone body, dissolve calcium hydroxide, and then permeate out, leading to a decrease in the compactness and strength of the hardened paste. Furthermore, the dissolution of calcium hydroxide leads to a decrease in the alkalinity of the matrix, which can further cause the decomposition of other hydration products.

Several comparative studies evaluated the water corrosion resistance of various grouts using strength loss, ion leaching, and microscopic analysis. Sun et al. [112] have analyzed the water corrosion resistance of different types of grouting materials, including singlefluid cement slurry, cement-water glass slurry, and industrial waste residue-based slurry. Zhao et al. [113] evaluated the water corrosion resistance characteristics of clay-cement paste slurry, clay-cement stabilized slurry, and single-fluid cement slurry by measuring the amount of calcium oxide dissolved in immersion water; they evaluated the water corrosion resistance characteristics of these three types of slurries. It was found that the viscosity of cement paste slurry exhibited the best water corrosion resistance. Weldes and Lange [114] conducted experimental research and evaluation using indexes such as the strength loss rate, sodium ion solidification rate, silicate dissolution amount, electrical conductivity, and total mass of dissolved matter. They compared these results with those of cement-based grout and discussed the reasons for the improved water corrosion resistance of industrial waste residue-based grout through microscopic tests. Gebauer and Coughlin [115] performed water corrosion resistance tests on hardened samples of different types of grouts, including pure cement grout, cement-water glass grout, cement mortar, and a newly developed grout. The above results suggested that industrial waste-based grouts and viscosity-modified grouts exhibited improved resistance.

## 3.3.3. Impermeability

Impermeability is an important performance index of the hardened stone body of grouting materials. It reflects how difficult it is for liquids or ions to infiltrate, diffuse, or migrate within the hardened stone body under pressure, chemical potential, or electric field. A grout with good impermeability usually has a relatively dense microstructure, making it resistant to penetration by external liquids and corrosive ions, which can lead to the deterioration of grouting material performance. Current studies analyzed the effects of density, mineral admixtures, mix proportions, and additives on the impermeability of specialized grout mixes, identifying the optimal parameters. Wang et al. [116] conducted research on the relationship between impermeability and density of ISW grouting materials. The results indicate that the density of polymer materials (non-aqueous reactive polymer grouting materials represented by polyurethane foam) is generally in the range of  $0.1-0.3 \text{ g/cm}^3$ , and the initial seepage pressure is in the range of 0.3-0.7 MPa, which can bear 30 to 70 m's head pressure when seepage grouting is carried out in hydraulic engineering. In practical applications, the amount of polymer grouting should be controlled, and the density of the polymer material should be adjusted according to the actual needs of different projects to meet the needs of seepage control in different projects. Mangat and Khatib [117] studied the influence of fly ash, silica fume, and slag powder on the impermeability of cement paste using a mortar permeameter. Their findings indicated that each material had different optimal replacement rates, which could optimize the impermeability of the hardened slurry. Shana and Jeffrey [118] analyzed the influencing factors of the permeability coefficient of slag cement slurry. They concluded that slag cement slurry prepared under suitable ratio conditions exhibited good impermeability and had broad application prospects. Zhang et al. [119] highlighted that alkali-activated cementitious materials prepared from dolomite, sodium carbonate, and calcined bentonite possess excellent water resistance. Zhao et al. [120] used the softening coefficient as a parameter to characterize the water resistance of the materials. The study found that incorporating 10–20% rice husk ash improved the water resistance of the prepared cementitious materials significantly. Yang et al. [121] discovered that after undergoing the same high-temperature test, the damage in terms of compressive strength and bond strength was lower in alkaliactivated mortar compared to cement mortar, indicating better high-temperature resistance. Li et al. [122] investigated the influence of the water-binder ratio and the ratio of liquid A to liquid B on the impermeability of hardened slurry samples during the preparation of cement-sodium silicate double slurry. The maximum impermeability pressure of the prepared samples reached 1.6 MPa. Porbaha et al. [123] developed a testing device for measuring the permeability coefficient of hardened grout to study the impermeability of clay-solidified grout. Song et al. [25] used a new additive, called Z201, to prepare a grouting material with excellent impermeability for shield tunnel walls. When the content of additive Z201 was 10%, the impermeability pressure reached 0.8 MPa.

## 3.3.4. Sulfate Attack Resistance

Groundwater sulfates can react with grout hydration products, causing expansion, strength loss, and cracks [124]. Groundwater used in grouting often contains sulfate substances such as sodium sulfate and magnesium sulfate. When the content of these sulfate substances is high, it can lead to the erosion and deterioration of the hardened paste over time, resulting in a decrease in its strength. Sulfate erosion can be categorized into internal erosion and external erosion based on the source of sulfate ions [125]. Internal erosion occurs when sulfate ions are present within the hardened paste itself, while external erosion is caused by sulfate ions from the surrounding environment. The process of external sulfate erosion can be divided into two stages: infiltration of sulfate ions into the matrix of the hardened paste from groundwater, which is influenced by the impermeability of

the hardened paste, and the reaction between sulfate ions and substances in the hardened paste matrix. The corrosion mechanism of sulfate on cement-sodium silicate grout was studied [32]. It was found that the gel products of traditional cement-based grout, mainly Ca(OH)<sub>2</sub> and C-S-H with a high Ca/Si ratio, react with sulfate ions to form expansive crystals, leading to the destruction of the matrix structure. Additionally, the stability of the C-S-H gel with a high Ca/Si ratio is poor, and the reaction consumption of Ca(OH)<sub>2</sub> reduces the alkalinity in the matrix, causing decomposition and destruction of the C-S-H gel. Carde et al. [126] identified calcium hydroxide and C-S-H as vulnerable and described degradation mechanisms. Yoshikawa and Toda [127] suggested that low concentrations of sulfate solutions can promote the early-stage strength development of hardened paste, while high concentrations of sulfate solutions can cause a decline in strength. Kourounis et al. [128] found that incorporating high-viscosity materials and steel slag into traditional cement-sodium silicate grout as partial replacements for cement can enhance the sulfate corrosion resistance of the grout. Therefore, it is important to conduct the appropriate research to develop grouting materials with improved sulfate corrosion resistance.

## 3.3.5. Freeze-Thaw Resistance

Karakurt and Bayazıt [129] compared the freeze–thaw resistance of alkali-activated concrete (with fly ash partially replacing kaolin) with ordinary Portland cement concrete. They found that ordinary Portland cement concrete reached a mass loss rate of 5% after 45 freeze–thaw cycles, while alkali-activated concrete required 98 freeze–thaw cycles to reach the same mass loss rate. Liu et al. [130] studied the effect of silica fume on the freeze–thaw resistance of alkali-activated fly ash/slag cementitious materials through experiments. The results showed that the addition of silica fume led to more damage and deterioration of the hardened samples under freeze–thaw cycles.

Aiken et al. [131] investigated the effects of freeze–thaw cycles on the mechanical properties of a grouting material composed of basalt fiber, slag powder, and fly ash. The results indicated that the material with a basalt fiber content of 0.18 vol% exhibited superior frost resistance, as well as the highest compressive and tensile strengths compared to other fiber contents under the same freeze–thaw cycles.

Hou et al. [132] emphasized that increasing the slag content in alkali-activated fly ash/slag cementitious materials can reduce the porosity and pore diameter of the hardened sample matrix, thereby enhancing the freeze–thaw resistance of the materials.

Vegas et al. [133] conducted 300 freeze–thaw cycle tests on alkali-activated fly ash mortar and traditional Portland cement-based mortar. It was concluded that alkali-activated materials exhibit better freeze–thaw resistance than traditional Portland cement.

Slavik et al. [134] investigated the freeze–thaw resistance of alkali-activated fluidized bed combustion bottom ash cementitious material. The results showed that the compressive strength of hardened samples retained over 80%, even after 50 freeze–thaw cycles, indicating good freeze–thaw resistance. These studies collectively suggest that alkali-activated ISW cement materials generally exhibit superior freeze–thaw resistance compared to ordinary Portland cement-based materials. The slag content, silica fume addition, and other compositions affect the performance of the grouting materials.

# 3.3.6. Corrosion Resistance

Several research studies suggested that alkali-activated materials, after hardening, exhibit high mechanical strength and corrosion resistance compared to traditional cementbased materials. Wu et al. [135] studied the high-temperature resistance of alkali-activated fly ash/slag-based cementitious materials. They noted that the critical temperature for the degradation of physical and mechanical properties of the alkali-activated materials prepared from fly ash/slag is between 400–600 °C. Kanaan [136] compared the sulfate attack resistance of alkali-activated high territory/slag-based paste materials with pure cement-based paste materials. The results showed that the paste samples containing slag exhibited the best sulfate attack resistance, followed by pure kaolin samples, while the cement-based samples had the least resistance to high temperatures. These findings suggest that alkali-activated materials, represented by ISW-based grouting materials, offer improved performance in terms of sulfate attack resistance, making them advantageous for various applications in construction and engineering. Fernández-Jiménez et al. [137] investigated the corrosion resistance of alkali-activated cementitious materials compared to cement-based cementitious materials. The results demonstrated that the corrosion resistance coefficient of alkali-activated cementitious materials was significantly higher than that of cement-based materials. Meanwhile, Xin et al. [138] found that under corrosive conditions, the appearance damage of cement-based samples was more severe compared to alkali-activated fly ash-based cementitious material samples. This can be attributed to the gel products in cement-based samples, primarily composed of Ca(OH)<sub>2</sub>, C-S-H gel, and C-A-H gel, with a high Ca/Si ratio. These gels are stable in alkaline environments but easily decompose and deteriorate in acidic environments. These findings suggest that alkali-activated cementitious materials possess superior corrosion resistance compared to cement-based materials, making them more suitable for applications where exposure to corrosive environments is a concern. From the above studies, it is easy to see that ISW grouting materials possess superior durability properties under specific conditions compared to ordinary materials. However, conventional modification methods can only improve a certain property, and it is necessary to find a convenient and efficient method to improve the overall durability of ISW grouting materials.

# 4. Challenges and Future Prospects

The rapid expansion of infrastructure has resulted in a global increase in tunnel construction. Concurrently, significant industrialization has generated substantial amounts of industrial solid waste (ISW), leading to resource wastage and environmental pollution. Utilizing ISW as grouting materials in tunnels offers an eco-friendly and cost-effective solution to address these interconnected issues. However, several key challenges must be addressed before the widespread application of ISW grouting materials can be achieved.

One primary challenge is ensuring a consistent high performance of ISW grouting materials. Due to the variability in ISW composition and properties, the mechanical strength, fluidity, impermeability, durability, and other engineering properties of ISW grouts can be unstable. Thorough characterization and optimal mix design are necessary to meet the demanding performance criteria. Research should systematically investigate the complex effects of different ISW types, proportions, chemical activators, and preparation processes on grout properties.

Another major concern is the environmental impact and ecological safety associated with the use of ISW grouting materials. While intended to mitigate pollution, the improper use of ISW with high levels of heavy metals or other hazardous substances can introduce new environmental risks. Comprehensive assessment frameworks must be established to evaluate the leaching potential and ecological effects of ISW grouts. Standardized procedures are required to treat ISW and purify grout mixes to comply with health and safety regulations. Further research should focus on developing high-quality grouting materials from industrial waste.

Furthermore, there is currently a lack of specific international codes, standards, and guidelines regarding ISW grout production and quality control. Extensive mechanical and durability testing needs to be conducted to determine reasonable technical specifications for engineering properties, mix proportions, construction methods, and acceptance criteria. Systematic field testing and monitoring in pilot projects are essential to validate ISW grout performance. The accumulated experience should be formulated into detailed protocols and regulations to ensure quality assurance and safe, widespread implementation.

Advanced microstructure analysis can provide valuable insights into the composition– structure–property relationships of ISW grouting materials, guiding the optimization of their engineering properties. Some key ways in which advanced microstructure analysis aids in this optimization include: (1) Scanning electron microscopy (SEM) coupled with energy-dispersive X-ray spectroscopy (EDS) can characterize the elemental distribution and chemical composition of different phases and reaction products in the grout's microstructure, revealing the role of specific ISW particles and hydration products in enhancing properties.

(2) Transmission electron microscopy (TEM) enables nanoscale examination of C-S-H gels and other nanoscale hydrates, facilitating an understanding of macro-scale strength and impermeability.

(3) Mercury intrusion porosimetry quantifies the pore size distribution, allowing for the study of pore parameters like total porosity, critical pore diameter, shape, and connectivity, which helps establish correlations between permeability and durability.

(4) XRD identifies crystalline reaction products, providing insights into strength development mechanisms by tracking formations such as calcium silicate hydrates and calcium aluminate hydrates at different curing ages.

(5) Fourier transform infrared spectroscopy detects chemical bonding information and can monitor the progress of pozzolanic reactions and hydration processes over time.

(6) Nuclear magnetic resonance spectroscopy analyzes silicate polymerization, revealing the cross-linking density of the C-S-H gel phase, which largely controls strength.

(7) Mass spectroscopy accurately identifies the ion types and concentrations leached from grout samples in durability testing, aiding in the evaluation of resistance mechanisms.

By correlating advanced microstructural observations with measured engineering properties, the optimal type and dosage of ISW can be determined to tailor grout behavior for specific applications. The micro–macro linkages provided by sophisticated analysis are invaluable for designing high-performance and durable ISW grouting materials.

Advanced techniques also offer opportunities to enhance ISW grouts. For example:

(1) Optimizing the particle size distribution through fine grinding and classification can improve particle packing density, rheology, and strength. Meanwhile, novel admixtures and nano-additives may regulate the setting time, reduce shrinkage, and enhance impermeability.

(2) Advanced sensors and monitoring technologies are utilized to monitor the rheological properties, uniformity, and curing process of the grouting process in real time for precise control and adjustment.

(3) Using 3D-printing technology, the geometry and microstructure of grouting materials can be precisely controlled to improve the mechanical properties of grouting and process efficiency.

(4) Microencapsulation technology encapsulates additives in tiny capsules, allowing for a precise controlled release effect that improves the performance and stability of the grouting material.

(5) Using computational simulation and numerical analysis methods, the grouting process can be accurately modeled and simulated to better understand material behavior and optimize the grouting design.

In the future, the integration of diverse industrial wastes into optimal grout design will be facilitated by advanced technical means, promoting sustainable infrastructure construction. More research should be directed towards the development of multifunctional grouts with self-sensing, crack-healing, and microbial properties for the creation of "smart tunnels". Resource utilization can be maximized through the recycling and reusing of excavated soil, demolition waste, and industrial waste. With persistent research efforts and interdisciplinary collaboration, the evolution of ISW grouting technology will benefit both the industry and the environment.

### 5. Conclusions

This comprehensive review of recent studies on utilizing industrial solid waste (ISW) in tunnel grouting materials demonstrates the promising potential of this sustainable approach. However, further research and development are needed to address the existing limitations and knowledge gaps. The main findings are as follows:

(1) The existing literature is well established on a variety of ISWs (including slag, fly ash, steel slag, red mud, silica fume, calcium carbide slag, etc.) as partial substitutes for cement in grout formulations. At the same time, depending on the ISW reactivity, ISW can be mixed up to 50%. Composite mineral admixtures can participate in the hydration reaction to produce strength-enhancing gel products, such as C-S-H and C-A-H. Therefore, although there is no uniform understanding of ISW as a grouting material, there is no doubt that the use of the right type of ISW in the right dosage can reduce the cost of grouting and provide an environmentally friendly recycling solution for the large quantities of ISW that are generated globally.

(2) The mechanical properties, fluidity, volume stability, impermeability, and durability of ISW grouts depend on the specific type and dosage of ISW. The development of strength relies on the pozzolanic reactivity of the ISW components. To achieve a balance between strength and workability loss, an optimal range of industrial solid waste (ISW) content is considered to be 20–50%. The particle size distribution and shape of ISW particles affect the rheology of the grout. It is crucial to optimize the mix design, considering the characteristics of ISW to meet the engineering requirements. Microstructure analysis reveals that ISW hydration and refinements in the micro-scale pore structure contribute to enhanced macro-scale properties.

(3) Before an extensive application of ISW grouts can take place, several technological barriers need to be addressed. The variability and uncertainty of ISW properties make it challenging to achieve consistent grout quality. There is a lack of standardized codes, specifications, and quality control procedures in this area. It is necessary to conduct rigorous assessments to evaluate the ecological impacts. The use of advanced computer simulation and 3D-printing techniques might enable the precise control of properties customized to local conditions.

In summary, the use of ISW as substitute raw materials in tunnel grouting offers significant potential for resource recycling and pollution mitigation. With ongoing research to address the current challenges, ISW grouting technology can be continually enhanced to deliver sustainable and high-performance solutions for underground construction globally. This review provides valuable insights to support further exploration in this emerging field.

**Author Contributions:** Conceptualization, B.J. and S.W.; methodology, M.W.; software, A.Z.; validation, S.H., S.W. and S.H.; formal analysis, S.H.; investigation, B.J. and M.W. and A.Z.; resources, M.W.; data curation, S.W. and A.Z.; writing—original draft preparation, B.J. and S.W.; writing—review and editing, B.J. and S.H.; visualization, M.W.; supervision, S.H.; project administration, A.Z.; funding acquisition, B.J. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by the Science and Technology Research Project of Chongqing Education Commission, grant number KJQN202205802, KJQN202303436, KJQN202305802; and the Scientific Research Project of Chongqing Vocational Institute of Engineering, grant number 2022KJB09.

Data Availability Statement: Not applicable.

**Conflicts of Interest:** The authors declare no conflict of interest.

#### References

- Guo, D.; Song, Z.; Xu, T.; Zhang, Y.; Ding, L. Coupling analysis of tunnel construction risk in complex geology and construction factors. J. Constr. Eng. Manag. 2022, 148, 04022097. [CrossRef]
- Huang, J.; Liu, N.; Ma, Z.; Lu, L.; Dang, K. The construction stability of large section tunnel considering the deterioration of clay mechanical properties. *Front. Mater.* 2023, 10, 1135276. [CrossRef]
- Zhang, M.; Yu, D.; Wang, T.; Xu, C. Coupling Analysis of Tunnel Construction Safety Risks Based on NK Model and SD Causality Diagram. *Buildings* 2023, 13, 1081. [CrossRef]
- 4. Qi, Q.; Nie, Y.; Wang, X.; Liu, S. Exploring the effects of size ratio and fine content on vibration compaction behaviors of gap-graded granular mixtures via calibrated DEM models. *Powder Technol.* **2023**, *415*, 118156. [CrossRef]
- Li, Z.; Zhao, J.; Hu, K.; Li, Y.; Liu, L. Adaptability Evaluation of Rotary Jet Grouting Pile Composite Foundation for Shallow Buried Collapsible Loess Tunnel. *Appl. Sci.* 2023, 13, 1570. [CrossRef]
- 6. Fan, H.; Xu, Q.; Lai, J.; Liu, T.; Zhu, Z.; Zhu, Y.; Gao, X. Stability of the loess tunnel foundation reinforced by jet grouting piles and the influence of reinforcement parameters. *Transp. Geotech.* **2023**, *40*, 100965. [CrossRef]

- He, Y.; Wang, H.; Zhou, J.; Su, H.; Luo, L.; Zhang, B. Water Inrush Mechanism and Treatment Measures in Huali Highway Banyanzi Tunnel—A Case Study. *Water* 2023, 15, 551. [CrossRef]
- Liu, S.; Nie, Y.; Hu, W.; Ashiru, M.; Li, Z.; Zuo, J. The influence of mixing degree between coarse and fine particles on the strength of offshore and coast foundations. *Sustainability* 2022, 14, 9177. [CrossRef]
- 9. Zhang, W.; Wang, Y.; Song, M.; Shen, Z. Industrial structure upgrading, technological innovation and comprehensive utilisation of solid waste. *Technol. Anal. Strateg. Manag.* 2023, 2217719. [CrossRef]
- 10. Huang, C.; Zhang, J.-H.; Zhang, A.-S.; Li, J.; Wang, X.-Y. Permanent deformation and prediction model of construction and demolition waste under repeated loading. *J. Cent. South Univ.* **2022**, *29*, 1363–1375. [CrossRef]
- 11. Fares, G.; Alsaif, A.; Alhozaimy, A. Hybridization and cost-performance analysis of waste tire steel fibers into high-volume powdered scoria rocks-based ultra-high performance concrete. *J. Build. Eng.* **2023**, *72*, 106568. [CrossRef]
- Siles-Castellano, A.B.; López-González, J.A.; Jurado, M.M.; Estrella-González, M.J.; Suárez-Estrella, F.; López, M.J. Compost quality and sanitation on industrial scale composting of municipal solid waste and sewage sludge. *Appl. Sci.* 2021, 11, 7525. [CrossRef]
- Ragupathi, V.; Panigrahi, P.; Nagarajan, G.S. Supercapacitor active material from Recycling. ECS J. Solid State Sci. Technol. 2023, 12, 024001. [CrossRef]
- 14. Simoni, S.; Corrêa, J.A.; Ribeiro, C.J.C.; Abrunhosa, F.A.; Chira, O.P.A. Evaluation of the contamination of the soil and water of an open dump in the Amazon Region, Brazil. *Environ. Earth Sci.* **2021**, *80*, 113.
- 15. Janas, M.; Zawadzka, A. Assessment of the monitoring of an industrial waste landfill. *Ecol. Chem. Eng. S* 2018, 25, 659–669. [CrossRef]
- 16. Wang, N.; Chai, X.; Guo, Z.; Guo, C.; Liu, J.; Zhang, J. Hierarchy performance assessment of industrial solid waste utilization— Tracking resource recycling and utilization centers in China. *Environ. Sci. Pollut. Res.* **2023**, *30*, 83330–83340. [CrossRef]
- 17. Willson-Levy, R.; Peled, A.; Klein-BenDavid, O.; Bar-Nes, G. Development of One-part geopolymers based on industrial carbonate waste. *Constr. Build. Mater.* 2023, 365, 130009. [CrossRef]
- 18. Zhang, L.; Jia, X.; Yang, G.; Wang, D. High-volume upcycling of waste sediment and enhancement mechanisms in blended grouting material. *Adv. Mater. Sci. Eng.* **2022**, 2022, 3225574. [CrossRef]
- 19. Liu, W.; Jiang, Y.; Zhao, Z.; Du, R.; Li, H. Material innovation and performance optimization of multi-solid waste-based composite grouting materials for semi-flexible pavements. *Case Stud. Constr. Mater.* **2022**, *17*, e01624. [CrossRef]
- 20. Wang, X.; Hu, J.; Zhang, X.; Wang, K.; Shen, D.; Liang, C. Experimental Study on Mineral Solid Waste Green Grouting Material Based on Electromagnetic Characteristic Detection. *Adv. Civ. Eng.* **2022**, 2022, 6261429. [CrossRef]
- 21. Yao, T. Study on the Effectiveness and Efficiency of Regulation on Resource Utilization of Industrial Solid Waste; Shanxi University of Finance and Economics: Taiyuan, China, 2021.
- 22. Zhang, L.; Jia, X.; Wang, C.; Jiang, Q.; Sun, C. Characterization of Magnesium Potassium Phosphate Cement-Based Grouting Material Blended with High Volume Industrial Wastes. *Adv. Mater. Sci. Eng.* **2022**, 2022, 7353985. [CrossRef]
- Dun, Z.; Wang, M.; Zou, Y.; Ren, L. Laboratory Tests on Performance of Waste-Clay-Brick-Powder Cement Grouting Materials for Ground Improvement in Mine Goaf. Adv. Eng. Mater. 2022, 24, 2101575. [CrossRef]
- 24. Lin, C.; Dai, W.; Li, Z.; Sha, F. Performance and microstructure of alkali-activated red mud-based grouting materials under class F fly ash amendment. *Indian Geotech. J.* 2020, *50*, 1048–1056. [CrossRef]
- Song, W.; Zhu, Z.; Pu, S.; Wan, Y.; Huo, W.; Song, S.; Zhang, J.; Yao, K.; Hu, L. Synthesis and characterization of eco-friendly alkali-activated industrial solid waste-based two-component backfilling grouts for shield tunnelling. *J. Clean. Prod.* 2020, 266, 121974. [CrossRef]
- Li, Y.; Debusschere, R.; Yuan, Q.; Li, J. Recycling of ground jet grouting waste as a supplementary cementitious material. *Resour. Conserv. Recycl.* 2023, 194, 106993. [CrossRef]
- 27. Shukla, N.; Harbola, M.; Sanjay, K.; Shekhar, R. Electrochemical fencing of Cr (VI) from industrial wastes to mitigate ground water contamination. *Trans. Indian Inst. Met.* **2017**, *70*, 511–518. [CrossRef]
- Yan, J.; Karlsson, A.; Zou, Z.; Dai, D.; Edlund, U. Contamination of heavy metals and metalloids in biomass and waste fuels: Comparative characterisation and trend estimation. *Sci. Total Environ.* 2020, 700, 134382. [CrossRef]
- 29. Ceballos, E.; Cama, J.; Soler, J.M.; Frei, R. Release and mobility of hexavalent chromium in contaminated soil with chemical factory waste: Experiments, Cr isotope analysis and reactive transport modeling. J. Hazard. Mater. 2023, 451, 131193. [CrossRef]
- 30. Zhang, X.; Li, W.; Tang, Z.; Wang, X.; Sheng, D. Sustainable regenerated binding materials (RBM) utilizing industrial solid wastes for soil and aggregate stabilization. *J. Clean. Prod.* **2020**, *275*, 122991. [CrossRef]
- 31. Zhang, C.; Fu, J.; Yang, J.; Ou, X.; Ye, X.; Zhang, Y. Formulation and performance of grouting materials for underwater shield tunnel construction in karst ground. *Constr. Build. Mater.* **2018**, *187*, 327–338. [CrossRef]
- 32. Yu, Z.; Yang, L.; Zhou, S.; Gong, Q.; Zhu, H. Durability of cement-sodium silicate grouts with a high water to binder ratio in marine environments. *Constr. Build. Mater.* **2018**, *189*, 550–559. [CrossRef]
- Zhang, G.; Liu, J.; Li, Y.; Liang, J. A pasty clay-cement grouting material for soft and loose ground under groundwater conditions. *Adv. Cem. Res.* 2017, 29, 54–62. [CrossRef]
- 34. Tian, K. Study and Application on High Property Grouting Material Used in Synchronous Grouting of Shield Tunnelling; Wuhan University of Technology: Wuhan, China, 2007.

- Zhang, G. Research and Application of New Industrial Residue Two-Shot Grouting Materials. Master's Thesis, Wuhan University of Technology, Wuhan, China, 2007.
- 36. Ding, Y. Experimental Study on Physical Mechanical Behavior of Backfill Grouting Material in Shield Tunneling. Master's Thesis, Southeast University, Nanjing, China, 2017.
- 37. Xu, K. The Research and Application on High-Performance Grouting Made by Shield Sediment; Wuhan University of Technology: Wuhan, China, 2011.
- Liang, X. Study of Material Performance and Material Ratio Design in Synchronized Grouting Technology for Shield Tunneling on Watery Strata; Chang'an University: Xi'an, China, 2009. (In Chinese)
- Chen, Y. Investigation into the Characteristics and Strength Simulation Test of the Complex Two-Component in Synchronous Grouting of Shield Tunneling. Master's Thesis, Beijing Jiaotong University, Beijing, China, 2017.
- 40. Yuyou, Y.; Zengdi, C.; Xiangqian, L.; Haijun, D. Development and materials characteristics of fly ash-slag-based grout for use in sulfate-rich environments. *Clean Technol. Environ. Policy* **2016**, *18*, 949–956. [CrossRef]
- 41. Wang, S.; Wang, J.-F.; Yuan, C.-P.; Chen, L.-Y.; Xu, S.-T.; Guo, K.-B. Development of the nano-composite cement: Application in regulating grouting in complex ground conditions. *J. Mt. Sci.* 2018, *15*, 1572–1584. [CrossRef]
- Li, Z.; Zhang, J.; Li, S.; Gao, Y.; Liu, C.; Qi, Y. Effect of different gypsums on the workability and mechanical properties of red mud-slag based grouting materials. J. Clean. Prod. 2020, 245, 118759. [CrossRef]
- 43. Lin, C.; Dai, W.; Li, Z.; Wang, Y. Study on the inorganic synthesis from recycled cement and solid waste gypsum system: Application in grouting materials. *Constr. Build. Mater.* **2020**, *251*, 118930. [CrossRef]
- Sha, F.; Jin, Q.; Liu, P. Development of effective microfine cement-based grouts (EMCG) for porous and fissured strata. *Constr. Build. Mater.* 2020, 262, 120775. [CrossRef]
- 45. Li, T.; Yue, Z.; Li, J.; Li, Q.; Li, Y.; Chen, G. Experimental study of improved cement silicate grouting material for broken surrounding rock. *J. Build. Eng.* **2023**, *74*, 106782. [CrossRef]
- 46. Zhang, C.; Fu, J.; Song, W. Mechanical model and strength development evolution of high content fly ash-cement grouting material. *Constr. Build. Mater.* **2023**, *398*, 132492. [CrossRef]
- 47. Shen, Q.; Jiang, R.; Cong, B.; Guo, B.; Shang, H.; Ji, X.; Li, L. Study on the effect of mineral admixtures on working and mechanical properties of the grouting material. *Front. Mater.* **2023**, *10*, 1197997. [CrossRef]
- Sun, X.; Miao, H.; Li, H.; Cui, W.; Zhang, F.; Liu, X.; Song, H.; Li, S. Study on the effect of mixed aggregate gradation on the performance of cement-based grouting material. *Concrete* 2019, *3*, 82–85.
- 49. Wang, M.; Wang, C.; Yu, J.; Li, Y.; Wen, P.; Fan, Q. Investigation of the grouting effect of blast furnace slag-based mortar on void road bases based on the grouting simulation test. *Constr. Build. Mater.* **2021**, *282*, 122567. [CrossRef]
- Ren, C.; Wang, W.; Li, G. Preparation of high-performance cementitious materials from industrial solid waste. *Constr. Build. Mater.* 2017, 152, 39–47. [CrossRef]
- Sha, F.; Li, H.; Pan, D.; Liu, H.; Zhang, X. Development of steel slag composite grouts for underground engineering. J. Mater. Res. Technol. 2020, 9, 2793–2809. [CrossRef]
- Mirza, J.; Mirza, M.; Roy, V.; Saleh, K. Basic rheological and mechanical properties of high-volume fly ash grouts. *Constr. Build. Mater.* 2002, 16, 353–363. [CrossRef]
- 53. Das, D.; Rout, P.K. Synthesis of Inorganic Polymeric Materials from Industrial Solid Waste. Silicon 2023, 15, 1771–1791. [CrossRef]
- 54. Lin, R.; Yang, L.; Pan, G.; Sun, Z.; Li, J. Properties of composite cement-sodium silicate grout mixed with sulphoaluminate cement and slag powder in flowing water. *Constr. Build. Mater.* **2021**, *308*, 125040. [CrossRef]
- Li, M.; Huang, G.; Cui, Y.; Wang, B.; Chang, B.; Yin, Q.; Zhang, S.; Wang, Q.; Feng, J.; Ge, M. Coagulation Mechanism and Compressive Strength Characteristics Analysis of High-Strength Alkali-Activated Slag Grouting Material. *Polymers* 2022, 14, 3980. [CrossRef]
- 56. Prentice, D.P.; Bernal, S.A.; Bankhead, M.; Hayes, M.; Provis, J.L. Phase evolution of slag-rich cementitious grouts for immobilisation of nuclear wastes. *Adv. Cem. Res.* 2018, *30*, 345–360. [CrossRef]
- 57. Li, S.; Sha, F.; Liu, R.; Zhang, Q.; Li, Z. Investigation on fundamental properties of microfine cement and cement-slag grouts. *Constr. Build. Mater.* **2017**, *153*, 965–974. [CrossRef]
- 58. Ghouleh, Z.; Guthrie, R.I.; Shao, Y. Production of carbonate aggregates using steel slag and carbon dioxide for carbon-negative concrete. *J. CO2 Util.* **2017**, *18*, 125–138. [CrossRef]
- 59. Li, Z.; Lin, C.; Sha, F.; Zhang, Q.; Zhang, J.; Qi, Y.; Gao, Y. Durability of a new type of cement-based composite grouting material under effects of chemical corrosion. *Mater. Express* **2020**, *10*, 948–954. [CrossRef]
- André, L.; Bacquié, C.; Comin, G.; Ploton, R.; Achard, D.; Frouin, L.; Cyr, M. Improvement of two-component grouts by the use of ground granulated blast furnace slag. *Tunn. Undergr. Space Technol.* 2022, 122, 104369. [CrossRef]
- 61. Gencel, O.; Karadag, O.; Oren, O.H.; Bilir, T. Steel slag and its applications in cement and concrete technology: A review. *Constr. Build. Mater.* **2021**, *283*, 122783. [CrossRef]
- 62. Sun, J.; Chen, Z. Effect of silicate modulus of water glass on the hydration of alkali-activated converter steel slag. *J. Therm. Anal. Calorim.* **2019**, *138*, 47–56. [CrossRef]
- 63. Yu, J.; Hua, S.; Qian, L.; Ren, X.; Zuo, J.; Zhang, Y. Development of steel slag-based solidification/stabilization materials for high moisture content soil. *J. Renew. Mater.* 2022, 10, 735.

- 64. Pei, S.-L.; Pan, S.-Y.; Li, Y.-M.; Chiang, P.-C. Environmental benefit assessment for the carbonation process of petroleum coke fly ash in a rotating packed bed. *Environ. Sci. Technol.* 2017, *51*, 10674–10681. [CrossRef]
- 65. Trimurtiningrum, R.; Ekaputri, J.J. Geopolymer grout material. Mater. Sci. Forum 2016, 841, 40–47. [CrossRef]
- Li, F.; Yang, Z.; Zhou, S.; Zhang, X.; Liu, D.; Su, Y.; Niu, Q. Fly-ash-based geopolymer modified by metakaolin for greener grouting material. Proc. Inst. Civ. Eng. Eng. Sustain. 2023, 40, 1–14. [CrossRef]
- 67. Huang, H.R.; Liao, Y.S.; Qunaynah, S.A.; Jiang, G.X.; Yuan, W.J. Effect of steel slag on shrinkage characteristics of calcium sulfoaluminate cement. *Mater. Sci. Forum* 2021, 1036, 263–276. [CrossRef]
- 68. Cui, P.; Wan, Y.; Shao, X.; Ling, X.; Zhao, L.; Gong, Y.; Zhu, C. Study on Shrinkage in Alkali-Activated Slag–Fly Ash Cementitious Materials. *Materials* **2023**, *16*, 3958. [CrossRef]
- 69. Deng, P.; Zheng, Z. Mechanical properties of one-part geopolymer masonry mortar using alkali-fused lead–zinc tailings. *Constr. Build. Mater.* **2023**, *369*, 130522. [CrossRef]
- 70. Zhang, J.; Guan, X.; Li, H.; Liu, X. Performance and hydration study of ultra-fine sulfoaluminate cement-based double liquid grouting material. *Constr. Build. Mater.* **2017**, *132*, 262–270. [CrossRef]
- Guo, L.; Zhou, M.; Wang, X.; Li, C.; Jia, H. Preparation of coal gangue-slag-fly ash geopolymer grouting materials. *Constr. Build. Mater.* 2022, 328, 126997. [CrossRef]
- Ruan, W.; Li, F.; Liao, J.; Gu, X.; Mo, J.; Shen, Y.; Zhu, Y.; Ma, X. Effects of water purifying material waste on properties and hydration mechanism of magnesium phosphate cement-based grouting materials. *Constr. Build. Mater.* 2022, 349, 128676. [CrossRef]
- Duan, D.; Wu, H.; Wei, F.; Song, H.; Ma, Z.; Chen, Z.; Cheng, F. Preparation, characterization, and rheological analysis of eco-friendly geopolymer grouting cementitious materials based on industrial solid wastes. *J. Build. Eng.* 2023, 78, 107451. [CrossRef]
- 74. Zhang, H.; Dong, J.; Wei, C. Material properties of cement doped with carbonated steel slag through the slurry carbonation process: Effect and quantitative model. *Metall. Mater. Trans. B* **2022**, *53*, 1681–1690. [CrossRef]
- 75. Cao, H.; Gao, Q.; Zhang, X.; Guo, B. Research progress and development direction of filling cementing materials for filling mining in iron mines of China. *Gels* **2022**, *8*, 192. [CrossRef]
- Xue, G.; Yilmaz, E.; Song, W.; Cao, S. Compressive strength characteristics of cemented tailings backfill with alkali-activated slag. *Appl. Sci.* 2018, *8*, 1537. [CrossRef]
- 77. Sha, F.; Liu, P.; Ding, Y. Application investigation of high-phosphorus steel slag in cementitious material and ordinary concrete. *J. Mater. Res. Technol.* **2021**, *11*, 2074–2091. [CrossRef]
- Deng, D.; Cao, G.; Liang, Y. Experimental Study on Lead-Smelting Slag as Paste Filling Cementing Material. *Geofluids* 2022, 2022, 6126881. [CrossRef]
- Li, J.-F.; Liu, Y.-T.; Li, S.-J.; Song, Y. Experimental investigation of synchronous grouting material prepared with different mineral admixtures. *Materials* 2022, 15, 1260. [CrossRef] [PubMed]
- 80. Lahalle, H.; Benavent, V.; Trincal, V.; Wattez, T.; Bucher, R.; Cyr, M. Robustness to water and temperature, and activation energies of metakaolin-based geopolymer and alkali-activated slag binders. *Constr. Build. Mater.* **2021**, *300*, 124066. [CrossRef]
- 81. Bai, T.; Liang, Y.; Li, C.; Jiang, X.; Li, Y.; Chen, A.; Wang, H.; Xu, F.; Peng, C. Application and validation of fly ash based geopolymer mortar as grouting material in porous asphalt concrete. *Constr. Build. Mater.* **2022**, *332*, 127154. [CrossRef]
- Wan, Y.; Hui, X.; He, X.; Xue, J.; Feng, D.; Chen, Z.; Li, J.; Liu, L.; Xue, Q. Utilization of flue gas desulfurization gypsum to produce green binder for dredged soil solidification: Strength, durability, and planting performance. J. Clean. Prod. 2022, 367, 133076. [CrossRef]
- 83. Sha, F.; Li, S.; Liu, R.; Li, Z.; Zhang, Q. Experimental study on performance of cement-based grouts admixed with fly ash, bentonite, superplasticizer and water glass. *Constr. Build. Mater.* **2018**, *161*, 282–291. [CrossRef]
- 84. Zhu, C.; Wan, Y.; Wang, L.; Ye, Y.; Yu, H.; Yang, J. Strength Characteristics and Microstructure Analysis of Alkali-Activated Slag–Fly Ash Cementitious Material. *Materials* **2022**, *15*, 6169. [CrossRef]
- Liu, J.-H.; Zhou, Z.-B.; Wu, A.-X.; Wang, Y.-M. Preparation and hydration mechanism of low concentration Bayer red mud filling materials. *Chin. J. Eng.* 2020, 42, 1457–1464.
- 86. Zhang, W.; Zhu, X.; Xu, S.; Wang, Z.; Li, W. Experimental study on properties of a new type of grouting material for the reinforcement of fractured seam floor. *J. Mater. Res. Technol.* **2019**, *8*, 5271–5282. [CrossRef]
- Carvalho, V.R.; Costa, L.C.B.; da Fonseca Elói, F.P.; da Silva Bezerra, A.C.; de Carvalho, J.M.F.; Peixoto, R.A.F. Performance of low-energy steel slag powders as supplementary cementitious materials. *Constr. Build. Mater.* 2023, 392, 131888. [CrossRef]
- 88. Li, C.; Hao, Y.F.; Zhao, F.Q. Preparation of fly ash-granulated blast furnace slag-carbide slag binder and application in total tailings paste backfill. *IOP Conf.* 2018, 322, 042003. [CrossRef]
- 89. Ehsani, A.; Nili, M.; Shaabani, K. Effect of nanosilica on the compressive strength development and water absorption properties of cement paste and concrete containing Fly Ash. *KSCE J. Civ. Eng.* **2017**, *21*, 1854–1865. [CrossRef]
- Rehman, S.; Iqbal, S.; Ali, A. Combined influence of glass powder and granular steel slag on fresh and mechanical properties of self-compacting concrete. *Constr. Build. Mater.* 2018, 178, 153–160. [CrossRef]
- 91. Xue, G.; Fu, Q.; Xu, S.; Li, J. Macroscopic mechanical properties and microstructure characteristics of steel slag fine aggregate concrete. *J. Build. Eng.* **2022**, *56*, 104742. [CrossRef]

- 92. Gopinathan, S.; Anand, K.B. Properties of cement grout modified with ultra-fine slag. *Front. Struct. Civ. Eng.* **2018**, *12*, 58–66. [CrossRef]
- Liu, J.; Guo, R. Applications of Steel Slag Powder and Steel Slag Aggregate in Ultra-High Performance Concrete. *Adv. Civ. Eng.* 2018, 2018, 1426037. [CrossRef]
- Wei, X.; Ni, W.; Zhang, S.; Wang, X.; Li, J.; Du, H. Influence of the key factors on the performance of steel slag-desulphurisation gypsum-based hydration-carbonation materials. *J. Build. Eng.* 2022, 45, 103591. [CrossRef]
- 95. Li, Y.; Min, X.; Ke, Y.; Liu, D.; Tang, C. Preparation of red mud-based geopolymer materials from MSWI fly ash and red mud by mechanical activation. *Waste Manag.* **2019**, *83*, 202–208. [CrossRef]
- Zhang, M.H. Microstructure, crack propagation, and mechanical properties of cement pastes containing high volumes of fly ashes. *Cem. Concr. Res.* 1995, 25, 1165–1178. [CrossRef]
- Liu, Z.; Ni, W.; Li, Y.; Ba, H.; Li, N.; Ju, Y.; Zhao, B.; Jia, G.; Hu, W. The mechanism of hydration reaction of granulated blast furnace slag-steel slag-refining slag-desulfurization gypsum-based clinker-free cementitious materials. *J. Build. Eng.* 2021, 44, 103289. [CrossRef]
- Liu, X.; Rong, Y.; Chen, X.; Chen, X.; Zhang, W. Recycling of Waste Stone Powder in High Fluidity Grouting Materials for Geotechnical Engineering Reinforcement. *Buildings* 2022, 12, 1887. [CrossRef]
- Chen, Z.; Qiao, J.; Yang, X.; Sun, Y.; Sun, D. A review of grouting materials for pouring semi-flexible pavement: Materials, design and performance. *Constr. Build. Mater.* 2023, 379, 131235. [CrossRef]
- 100. Li, S.; Zhang, J.; Li, Z.; Liu, C.; Chen, J. Feasibility study on grouting material prepared from red mud and metallurgical wastewater based on synergistic theory. *J. Hazard. Mater.* **2021**, 407, 124358. [CrossRef] [PubMed]
- 101. Moghadam, M.J.; Ajalloeian, R.; Hajiannia, A. Preparation and application of alkali-activated materials based on waste glass and coal gangue: A review. *Constr. Build. Mater.* **2019**, 221, 84–98. [CrossRef]
- Xu, J.; Kang, A.; Wu, Z.; Xiao, P.; Gong, Y. Effect of high-calcium basalt fiber on the workability, mechanical properties and microstructure of slag-fly ash geopolymer grouting material. *Constr. Build. Mater.* 2021, 302, 124089. [CrossRef]
- 103. ElKhatib, L.W.; Elkordi, A.; Khatib, J. Methods and surface materials repair for concrete structures—A review. *BAU J. Sci. Technol.* **2023**, *4*, 7. [CrossRef]
- Xiang, J.; Liu, L.; Cui, X.; He, Y.; Zheng, G.; Shi, C. Effect of limestone on rheological, shrinkage and mechanical properties of alkali—Activated slag/fly ash grouting materials. *Constr. Build. Mater.* 2018, 191, 1285–1292. [CrossRef]
- Wu, H.; Huang, B.; Shu, X.; Yin, J. Utilization of solid wastes/byproducts from paper mills in Controlled Low Strength Material (CLSM). Constr. Build. Mater. 2016, 118, 155–163. [CrossRef]
- He, J.; Bu, X.; Bai, W.; Zheng, W.; Gao, Q.; Wang, Y. Preparation and properties of self-compacting alkali-activated slag repair mortar. *Constr. Build. Mater.* 2020, 252, 119034. [CrossRef]
- 107. Li, X.; Hao, J. New preparation of super-early-strength grouting materials by ternary complex system. *Adv. Cem. Res.* **2018**, *30*, 139–147. [CrossRef]
- Borçato, A.G.; Thiesen, M.; Medeiros-Junior, R.A. Mechanical properties of metakaolin-based geopolymers modified with different contents of quarry dust waste. *Constr. Build. Mater.* 2023, 400, 132854. [CrossRef]
- 109. Li, M.; Gu, K.; Chen, B. Effects of flue gas desulfurization gypsum incorporation and curing temperatures on magnesium oxysulfate cement. *Constr. Build. Mater.* 2022, 349, 128718. [CrossRef]
- Permeh, S.; Lau, K. Review of Electrochemical Testing to Assess Corrosion of Post-Tensioned Tendons with Segregated Grout. Constr. Mater. 2022, 2, 70–84. [CrossRef]
- Bentz, D.P.; Garboczi, E.J. Modelling the leaching of calcium hydroxide from cement paste: Effects on pore space percolation and diffusivity. *Mater. Struct.* 1992, 25, 523–533. [CrossRef]
- 112. Sun, C.; Hui, R.; Qu, W.; Yick, S. Progress in corrosion resistant materials for supercritical water reactors. *Corros. Sci.* 2009, *51*, 2508–2523. [CrossRef]
- 113. Zhao, Y.; Taheri, A.; Karakus, M.; Chen, Z.; Deng, A. Effects of water content, water type and temperature on the rheological behaviour of slag-cement and fly ash-cement paste backfill. *Int. J. Min. Sci. Technol.* **2020**, *30*, 271–278. [CrossRef]
- 114. Weldes, H.H.; Lange, K.R. Properties of soluble silicates. Ind. Eng. Chem. 1969, 61, 29–44. [CrossRef]
- 115. Gebauer, J.; Coughlin, R.W. Preparation, properties & corrosion resistance of composites of cement mortar and organic polymers. *Cem. Concr. Res.* **1971**, *1*, 187–210.
- Wang, F.M.; Shi, M.S.; Li, H.J.; Zhong, Y.H. Experimental Study on the Anti-Permeability Properties of Polymer Grouting Materials. *Adv. Mater. Res.* 2011, 284–286, 1952–1955. [CrossRef]
- 117. Mangat, P.S.; Khatib, J.M. Influence of Fly Ash, Silica Fume, and Slag on Sulfate Resistance of Concrete. *ACI Mater. J.* **1993**, *92*, 542–552.
- 118. Opdyke, S.M.; Evans, J.C. Slag-cement-bentonite slurry walls. J. Geotech. Geoenviron. Eng. 2005, 131, 673–681. [CrossRef]
- Zhang, S.; Wang, C.; Gao, G. Study of the Mechanical Properties of Alkali-Activated Solid Waste Cementitious Materials at the Interface. ACS Omega 2022, 7, 26531–26536. [CrossRef] [PubMed]
- Zhao, F.-Q.; Liu, H.-J.; Hao, L.-X.; Li, Q. Water resistant block from desulfurization gypsum. Constr. Build. Mater. 2012, 27, 531–533.
   [CrossRef]
- Yang, K.-H.; Song, J.-K.; Lee, J.-S. Properties of alkali-activated mortar and concrete using lightweight aggregates. *Mater. Struct.* 2010, 43, 403–416. [CrossRef]

- 122. Li, S.; Zhang, J.; Li, Z.; Gao, Y.; Qi, Y.; Li, H.; Zhang, Q. Investigation and practical application of a new cementitious anti-washout grouting material. *Constr. Build. Mater.* 2019, 224, 66–77. [CrossRef]
- 123. Porbaha, A.; Shibuya, S.; Kishida, T. State of the art in deep mixing technology. Part III: Geomaterial characterization. *Proc. Inst. Civ. Eng. Ground Improv.* 2000, *4*, 91–110. [CrossRef]
- 124. Ruan, W.; Liao, J.; Mo, J.; Li, F.; Gu, X.; Ma, Y.; Zhu, Y.; Ma, X. Effects of red mud on properties of magnesium phosphate cement-based grouting material and its bonding mechanism with coal rock. *Ceram. Int.* **2023**, *49*, 2015–2025. [CrossRef]
- 125. Sha, F.; Fan, R.; Gu, S.; Xi, M. Strengthening effect of sulphoaluminate cementitious grouting material for water-bearing broken rocky stratum. *Constr. Build. Mater.* **2023**, *368*, 130390. [CrossRef]
- 126. Carde, C.; François, R.; Torrenti, J.-M. Leaching of both calcium hydroxide and CSH from cement paste: Modeling the mechanical behavior. *Cem. Concr. Res.* **1996**, *26*, 1257–1268. [CrossRef]
- 127. Yoshikawa, M.; Toda, T. In vivo estimation of periapical bone reconstruction by chondroitin sulfate in calcium phosphate cement. *J. Eur. Ceram. Soc.* 2004, 24, 521–531. [CrossRef]
- 128. Kourounis, S.; Tsivilis, S.; Tsakiridis, P.E.; Papadimitriou, G.D.; Tsibouki, Z. Properties and hydration of blended cements with steelmaking slag. *Cem. Concr. Res.* 2007, *37*, 815–822. [CrossRef]
- 129. Karakurt, C.; Bayazıt, Y. Freeze-Thaw Resistance of Normal and High Strength Concretes Produced with Fly Ash and Silica Fume. *Adv. Mater. Sci. Eng.* **2015**, 2015, 830984. [CrossRef]
- 130. Liu, Y.; Zhou, X.; Lv, C.; Yang, Y.; Liu, T. Use of Silica Fume and GGBS to Improve Frost Resistance of ECC with High-Volume Fly Ash. *Adv. Civ. Eng.* **2018**, 2018, 7987589. [CrossRef]
- Aiken, T.A.; Kwasny, J.; Sha, W.; Tong, K.T. Mechanical and durability properties of alkali-activated fly ash concrete with increasing slag content. *Constr. Build. Mater.* 2021, 301, 124330. [CrossRef]
- 132. Hou, D.; Li, D.; Hua, P.; Jiang, J.; Zhang, G. Statistical modelling of compressive strength controlled by porosity and pore size distribution for cementitious materials. *Cem. Concr. Compos.* **2019**, *96*, 11–20. [CrossRef]
- Vegas, I.; Urreta, J.; Frías, M.; García, R. Freeze–thaw resistance of blended cements containing calcined paper sludge. *Constr. Build. Mater.* 2009, 23, 2862–2868. [CrossRef]
- Slavik, R.; Bednarik, V.; Vondruska, M.; Nemec, A. Preparation of geopolymer from fluidized bed combustion bottom ash. J. Mater. Process. Technol. 2008, 200, 265–270. [CrossRef]
- 135. Wu, W.; Ma, S.; Wang, Y.; Wang, X.; Xu, L.; Ye, S. Evaluation of Sulfate Resistance of One-Part Alkali-Activated Materials Prepared by Mechanochemistry. *ACI Mater. J.* **2023**, *120*, 37–51.
- 136. Kanaan, D. Developing of One-Part Alkali-Activated Self-Consolidating Concrete and Investigating Its Sulfate Attack Resistance; Concordia University: Montreal, QC, Canada, 2021.
- Fernández-Jiménez, A.; García-Lodeiro, I.; Palomo, A. Durability of alkali-activated fly ash cementitious materials. *J. Mater. Sci.* 2007, 42, 3055–3065. [CrossRef]
- 138. Luo, X.; Xu, J.; Bai, E.; Li, W. Systematic study on the basic characteristics of alkali-activated slag-fly ash cementitious material system. *Constr. Build. Mater.* **2012**, *29*, 482–486. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.